Author's response to reviews

Title: Wound healing effects of Malaysian propolis and Brazilian red propolis on connective tissue fibroblasts in the healing process

Authors:

ANN JACOB (annjacob.aj90@gmail.com)
Abhishek Parolia (abhishek_parolia@imu.edu.my)
ALLAN PAU (allan_pau@imu.edu.my)
Fabian Davamani (fabian_davamani@imu.edu.my)

Version: 6
Date: 10 August 2015

Author's response to reviews: see over
Dear Editor in Chief,

I would like to thank both reviewers for their precious time devoted in the review process of my manuscript and their suggestions to improve this manuscript.

<table>
<thead>
<tr>
<th>Reviewer’s comment</th>
<th>Author’s reply</th>
</tr>
</thead>
<tbody>
<tr>
<td>Thank you for arranging emails from manuscript authors confirming the change of authorship on your manuscript. However, I note that we are still awaiting confirmation from author IC Porto. This is particularly important, as he was one of the authors removed from your manuscript during the review process.</td>
<td><strong>All done</strong></td>
</tr>
<tr>
<td>- We note that a final remaining point of referee has not been addressed. Specifically, the comment which stated 'Please note in Figure 4A, the figures given in the Y-axis have commas. Please change the commas to dots (periods).' We would therefore ask you to please address this now.</td>
<td></td>
</tr>
<tr>
<td>- We must again ask you to please remove your Title Page from your additional files. It is only necessary that this page be included at the beginning of your manuscript document.</td>
<td></td>
</tr>
</tbody>
</table>

I hope, we could manage to answer all queries raised after the review process of this manuscript! I will be looking forward to your kind reply.

Regards,

Dr Abhishek Parolia